Author’s response to reviews

Title: Expression of nm23-H1 gene product in esophageal squamous cell carcinoma and its association with vessel invasion and survival

Authors:

Masaki Tomita (mtomita@post.miyazaki-med.ac.jp)
Takanori Ayabe (ayabe@fc.miyazaki-med.ac.jp)
Yasunori Matsuzaki (matuzaki@fc.miyazaki-med.ac.jp)
Masao Edagawa (meda@fc.miyazaki-med.ac.jp)
Masayuki Maeda (MASA@post.miyazaki-med.ac.jp)
Tetsuya Shimizu (kairiku@fc.miyazaki-med.ac.jp)
Masaki Hara (haramasa@fc.miyazaki-med.ac.jp)
Toshio Onitsuka (ono@fc.miyazaki-med.ac.jp)

Version: 4 Date: 18 Apr 2001

PDF covering letter
Dear Sir:

I would like to resubmit the enclosed, revised manuscript entitled “Expression of nm23-H1 gene product in esophageal squamous cell carcinoma and its association with vessel invasion and survival” by M. Tomita et. al. for publication in ‘BMC Cancer’.

We have modified the title and text of our manuscript in accordance with the reviewers' comments. We hope these changes will make our manuscript acceptable for ‘BMC Cancer’.

Please forward any correspondence regarding this manuscript to me.

Thank you very much for your kind consideration.

Sincerely yours,
Masaki Tomita, M. D. and Ph. D.
Department of Surgery II,
Miyazaki Medical College,
Kiyotake, Miyazaki, 889-1692, Japan
Tel: 81-985-85-2291; Fax: 81-985-85-5563
e-mail:mtomita@post.miyazaki-med.ac.jp
Response to prof. Tim Kute

I find the paper much improved and acceptable at this point. I do however note that none of my comments were addressed in the revision. Since my comments will be available for the reader, I am happy to let this problem be ignored.
I do not have any competing interests or any other gains concerning this paper.

Thank you very much for your kind consideration.

In our revised text, we did not ignore the reviewer’s comments. We have no intention of ignoring reviewer’s comments.
Response to prof. Sung Hoon Noh

In the revised manuscript, authors assessed the data adequately. And authors also analyzed the survival with the patients enrolled in this study.
I do agree on publishing the above article.
However, some editing is necessary.

Thank you very much for your kind consideration.

I am happy to read the reviewer’s comment: “I do agree on publishing the above article”.
In our revised text, we have modified the title and text of our manuscript.